Author's response to reviews

Title: Dissemination of multidrug-resistant tuberculosis in a patient with acute HIV infection

Authors:

Kyung Mo Yoo (kyungmo.yoo@samsung.com)
Eun-Jeong Joo (eunjeong.joo@samsung.com)
Joon-Sup Yeom (joonsup.yeom@samsung.com)
Seoung-Wan Chae (swan.chae@samsung.com)
Shin Yeoung Lee (shinyeong10.lee@samsung.com)
Ki Joong Han (imdr.han@samsung.com)

Version: 2 Date: 2 April 2014

Author's response to reviews: see over
Dear Editor,

Enclosed for your consideration for publication in **BMC Infectious Diseases** is the manuscript entitled “**Dissemination of multidrug-resistant tuberculosis in a patient with acute HIV infection**”.

We would be grateful if the manuscript could be reviewed again and considered for publication in the journal **BMC Infectious Diseases**.

I certify that all authors concurred with the submission and that the material submitted for publication has not been previously reported and is not under consideration for publication elsewhere.

Sincerely yours

Eun-Jeong Joo, M.D. (Corresponding Author)
Department of Internal Medicine,
Kangbuk Samsung Hospital, Sungkyunkwan University School of Medicine, 29 Saemunan-ro, Jongno-gu, Seoul, 110-746, South Korea
Tel: +82-2-2001-8533 / Fax: +82-2-2001-1596
E-mail: eunjeong.joo@samsung.com or iamjoo@hanmail.net